

Check for updates

Research Article

Interventions that challenge established and accepted clinical practice: lessons learnt from a process evaluation of the STOP-APE trial

Agnieszka Ignatowicz[®],^{1*} Sheila Greenfield[®],¹ Pooja Gaddu[®],² Clare Prince,³ Mark Toshner[®],⁴ Graham Robinson[®],⁵ Jonathan Rodrigues[®],⁵ Susan Jowett[®],¹ Simon Noble[®],⁶ Michael Newnham[®],¹ Alice Turner[®] and Daniel Lasserson[®]

Published February 2025 DOI: 10.3310/PSDG7298

Abstract

Background: Developing and implementing interventions that change clinical practice can be challenging and complex. Such interventions can be particularly difficult when attempting to change established behaviours and practices. While extensive literature on implementation of interventions that focus on changing clinical practice exists, understanding of the difficulties involved in implementing interventions that go against accepted clinical practice is limited.

Objectives: To describe the challenges involved in delivering a complex intervention that goes against established clinical practice, using a clinical trial assessing the balance of benefits and risks of withholding anticoagulation for subsegmental pulmonary embolism as an example.

Design and methods: This study draws from a process evaluation conducted as part of a clinical trial. The evaluation utilised semistructured interviews with patients and clinicians during the trial's internal pilot phase to investigate the acceptability of withholding anticoagulant medication and participants' experiences within the trial. The data were analysed using the framework method.

Setting and participants: Eight patients with subsegmental pulmonary embolism (six females and two males) and three acute care clinicians (two males and one female) from five trial sites were interviewed.

Results: Our findings indicated that factors such as clinician equipoise, discomfort with certain patient characteristics, and effective patient communication are closely connected and significantly impact both the process of changing clinical practice and the conduct of the trial. Clinicians faced difficulties in approaching eligible patients for trial participation, especially when a diagnosis and treatment plan had already been provided by another clinician. The tension between maintaining clinical equipoise and addressing the needs of unwell patients further complicated decisions, particularly when withholding anticoagulation in those with severe symptoms or multiple comorbidities. Communication about the risks and benefits of non-medication strategies for pulmonary embolism was also challenging, with concerns about undermining patient trust. Patients, on the other hand, expressed considerable anxiety about not receiving anticoagulants, with their perspectives on study participation and treatment heavily influenced by their prior health experiences and ongoing medical conditions. The active involvement of clinicians in the consent process had a positive effect on patients' perceptions and experiences, with many feeling reassured in knowing they could contact clinical staff if needed.

Institute of Applied Health Research, University of Birmingham, Edgbaston, Birmingham, UK

²Clinical Trials Unit, University of Birmingham, Edgbaston, Birmingham, UK

³Thrombosis UK, Llanwrda, Wales, UK

⁴Heart & Lung Research Institute, University of Cambridge, Cambridge, UK

⁵Department of Radiology, Royal United Hospitals Bath NHS Trust, Bath, UK

⁶Marie Curie Palliative Care Research Group, University of Cardiff, Cardiff, UK

⁷Health Sciences Division, University of Warwick, Coventry, UK

^{*}Corresponding author a.m.ignatowicz@bham.ac.uk

Limitations: Small sample size of patients and clinicians across limited study sites; single method of data collection. **Conclusions:** Our results highlight the multifaceted challenges when attempting to conduct studies that challenge accepted practices and norms. These complexities are deeply intertwined, influencing both clinical decision-making and patient recruitment for those studies.

Future work: Future research should focus on developing strategies to help clinicians maintain equipoise and communicate the risks and benefits of interventions, while also deepening the understanding of patients' experiences and perceptions to enhance recruitment strategies.

Ethical approval: Wales REC 6, Reference: 20/WA/0256, approved 30 September 2020.

Funding: This article presents independent research funded by the National Institute for Health and Care Research (NIHR) Health Technology Assessment programme as award number NIHR128073.

A plain language summary of this research article is available on the NIHR Journals Library Website https://doi. org/10.3310/PSDG7298.

Background

Developing and implementing interventions that change clinical practice can be complex and challenging. Such interventions are particularly difficult when attempting to change behaviours and practices since they necessitate clinicians learning new behaviours while unlearning old ones.^{2,3} Other factors can add to this inherent difficulty, such as clinicians' beliefs about the appropriateness of the intervention, and time and resources needed to implement the change.^{4,5} While there exists extensive literature on drivers and barriers to implementation of interventions that focus on changing clinical practice,6-8 understanding of the difficulties involved in implementing interventions that go against established and accepted clinical practice is limited.9 The clinicians not only have a difficult time adopting new behaviours and practices but also struggle with the social and ethical implications associated with these new practices. 10,11 Patient preferences and the dynamics of the professional-patient relationship can also play a role.

In this article, we use the example of a trial testing the balance of benefits and risks of withholding anticoagulation for isolated or incidental subsegmental pulmonary embolism (PE), a treatment that otherwise is accepted as standard care. We describe the challenges experienced when implementing such an intervention. Lessons learnt presented here are drawn from a qualitative process evaluation undertaken as part of the STOP-APE trial.¹² The evaluation investigated, during the internal pilot phase of the trial, the acceptability of not providing anticoagulant medication and clinician and patient experiences in the study.

Context of the current study

Pulmonary embolism is a potentially serious condition, whereby blood clots cause a blockage of the blood supply to the lungs.¹³ The symptoms of a PE depend on the size and location of the blood clot. The established

treatment for PE includes anticoagulant medications that are taken over at least 3 months. As the scanning technology for PE has become more sensitive, smaller clots are being diagnosed. However, small PEs may not cause any symptoms and may be found incidentally on scans performed for other reasons. In these situations, it is unclear whether treatment is required as these clots in smaller blood vessels away from the centre of the lungs (called subsegmental PE) may be removed by the body's own mechanisms without needing medications. At the same time, anticoagulant medication can cause side effects in some patients, such as bleeding and severe bruising, which, if severe, may need hospital treatment. 15

The STOP-APE trial was designed to test how a strategy of withholding anticoagulation in subsegmental PE compares to standard care which is full anticoagulation for at least 3 months. It aimed to recruit 1466 consenting adult patients with subsegmental PE from approximately 50 trial sites in secondary care clinical settings of emergency departments, ambulatory care and acute medical units within NHS hospitals in the UK. The internal pilot study begun in 2020, with a nested process evaluation to evaluate the acceptability of the intervention. However, due to restrictions imposed by the COVID-19 pandemic, the trial could not recruit for 7 months. When the recruitment restarted, we encountered various difficulties that generated uncertainty about the trial progress. Subsequently, with the advice from the funder, the trial was closed prematurely in December 2022.

There are two linked papers to this study in the National Institute for Health and Care Research (NIHR) Library: the protocol¹² and the synopsis that details the challenges and reasons why the STOP-APE trial was halted prematurely.¹² This qualitative article draws on our learning from the interviews with patients and clinicians and explores difficulties involved in implementing intervention (withholding anticoagulation in subsegmental PE) that is a marked change from standard practice.

Methods

Details of the main study design, sampling, data collection and analysis are published in the protocol. The embedded process evaluation employed semistructured interviews and aimed to interview up to 30 patients and 30 healthcare professionals taking part in the trial to allow for data saturation. Within our samples, we aimed for maximum variation to include the range of characteristics of eligible participants (e.g. site and symptomatic/incidental patients).

Data collection took place at five NHS sites throughout the UK between April and September 2021. All interviews were conducted by an experienced qualitative researcher (AI, female and social scientist) and were audio-recorded. The interviews lasted between 24 and 55 minutes and were undertaken via Zoom (Zoom Video Communications, San Jose, CA, USA) or telephone. The topic guide explored attitudes and practical issues surrounding patient understanding of PE and its management, tolerance of risk by patients and clinicians, preferences for content and delivery of information and any potential concerns. Verbal informed consent was obtained from all study participants.

Data collection and analysis were undertaken concurrently. Gale et al.'s framework method¹⁸ was adopted to analyse the data with the aid of the NVivo program (QSR International, Warrington, UK) for data management. First, Al read three interview transcripts to identify examples of participants' perceptions of and experiences in the study; views on acceptability of a no anticoagulation treatment strategy for isolated subsegmental PE; understanding of risk and incidental diagnosis; and any potential concerns. The codes were developed both 'horizontally' (by coding each interview as a standalone source of data) and 'vertically' (by scanning across the interviews for the specific areas described above). This created an initial set of codes and categories, and all the remaining transcripts were coded against that coding framework. The initial results of the analysis were then discussed during a study meeting with three members of the analysis team (AI, SG and DS). Following agreement on the initial themes, as our final analytic step, Al mapped the themes that emerged from the coding process to two broader themes that were related to practical and methodological challenges experienced by the team when designing and implementing the trial.

Patient and public involvement

We worked with patient partners with lived experience of thrombosis as we designed and delivered the trial, including one funding co-applicant with lived experience of thrombosis. Patients were members of our Trial Management Group and have advised us throughout on trial set-up, patient leaflet design and wording, mechanisms to increase recruitment, interpretation of our qualitative data and ultimately supported the decision of premature cessation of the trial.

Results

Eight patients (six females and two males) and three acute care clinicians (two males and one female) were interviewed due to the trial stopping prematurely (*Table 1*). All of the participants approached for an interview accepted the invitation.

The results were synthesised under three inter-related themes, which were presented separately for clarity: (1) approaching patients to take part in the trial; (2) recruiting and treating patients who are unwell; and (3) navigating communication about non-medication and patient preference for the trial arm.

Approaching patients to take part in the trial

Clinicians talked about instances where eligible patients were not approached to take part in the study because other clinicians who saw patients first believed they needed anticoagulants. Such instances led the clinicians to reflect on the broader challenges of changing current

TABLE 1 Participant characteristics

Participant group	Variables	Categories	Count
Patients	Gender	Female	6
		Male	2
	Trial recruitment site	Site 1	2
		Site 2	3
		Site 3	2
		Site 4	1
	Randomised clinical trial arm	Anticoagulants	5
		No anticoagulants	3
Clinicians	Gender	Female	1
		Male	2
	Trial recruitment site	Site 1	1
		Site 3	1
		Site 5	1

practice and the perceived appropriateness of approaching patients who were already diagnosed and informed about the treatment. For example, one clinician recalled a situation where the patient ended up feeling conflicted because the initial treatment plan was challenged during a conversation about taking part in the study:

Someone clearly told them [the patient] that this is what they need, and now you're saying that maybe that's not the case, and certainly [with] one patient it felt a little bit like if I carried on the conversation for too long, there was going to introduce the trust issue. That they would be like: well, hang about, you're telling me this isn't clear cut but, you know, it was absolutely clear cut in their mind. And they [the patient] started to get a little bit nervous and confused. And I didn't push, you know, I thought they would lose trust in us as doctors.

Clinician

These decisions were often influenced by whether the patient had already received information about their diagnosis from another clinician. One clinician reflected:

I honestly think that the key thing was that, it was the fact that they their first conversation after the diagnosis was with someone who knew about the trial and was interested in enrolling them versus someone who believes that the treatment was absolute and there were no other options, and I think ... I mean this gentleman [the patient], he was gonna say no ... the fact he had, I thought, there were a number of co-morbidities and because this patient maybe was relatively well ... I didn't bother approaching him.

Clinician 2

Another clinician echoed this sentiment, noting that patients were sometimes not put forward for the trial because the treating clinician was convinced that anticoagulation was necessary:

We've had one patient whose clinician didn't want them to be treated, they didn't want them treated regardless, and so they didn't. I wanted to put them forward for the trial but they weren't prepared for them to be anticoagulated.

Clinician 3

Adding to these challenges, clinicians highlighted a general lack of awareness about the trial among their colleagues, which further hindered patient identification and recruitment:

There are very few people in this [NHS] trust, who are trying to recruit and enrol, we're trying to work on that, but it's not happening at the moment. And the vast majority of other people just don't have any idea what's happening or aren't aware enough and so their standard presentation of a blood clot is the same as it's always been, which is you've got a blood clot gone to your lungs that serious, you need treatment and you need at least three months of treatment and so here's the tablet and off you go.

Clinician 1

Recruiting and treating patients who are unwell

Clinicians described a tension between the need to balance the risks and benefits of treating with anticoagulants for patients who were visibly unwell. Although clinicians acknowledged the clinical equipoise around the idea of treating or not treating subsegmental PE with anticoagulants, they generally held a view that there may be some patients who will still require medication and therefore may not be the best candidate for the trial. For instance, one clinician expressed the difficulty in withholding anticoagulants for a recruited patient who was unwell:

The only thing is symptomatic patients. Though the one patient I have recruited if he wasn't in the placebo group, I would have treated him. Later down the line when I actually met him because he was bizarrely symptomatic, acutely breathless when his sub-segmental PE was identified. And so it's harder to not treat that population, normally sub-segmentals you treat are, it was never a PE in the first place and they weren't symptomatic from it. Whereas his story was actually, he seemed to be very symptomatic from what was only a subsegmental PE. But luckily, he was put in the treatment arm.

Clinician 1

Another clinician shared a similar concern, highlighting specific cases where withholding treatment might not be appropriate:

I wouldn't do it with someone with a DVT, cancer. I mean I suppose the only thing is if they had limited reserve so if they had already for whatever reason cardio respiratory reasons that a small pulmonary vascular event could cause significant harm, that would be a reason that I would not want to not treat

Clinician 3

In contrast, another clinician noted that in many cases, the subsegmental PE was an incidental finding and not

associated with significant symptoms, which made the decision to not being anticoagulated more straightforward:

In terms of the sub-segmental PE, symptoms have not been an issue because they're often, they're normally an incidental sub-segmental finding.

Clinician 2

However, the challenge intensified when patients had multiple comorbidities or were generally unwell, making the decision to withhold anticoagulation even more complex:

The patient we didn't enrol had quite a lot of co-morbidities and had quite a lot of other health issues and a lot of contact with medic medical teams ... So, you know, this gentlemen was, he was unwell, really unwell.

Clinician 3

Navigating communication about non-medication and patient preference for the trial arm.

Clinicians reported that they frequently had to take a different approach when discussing non-medication with prospective trial participants. They emphasised the importance of clear communication to ensure patients understood that, in some cases, not medicating might be the appropriate course of action:

I tend to ask them [the patient] what they know about what they've come in with and what they've been diagnosed with or what you know, what they're in ambulatory care with. And then obviously you get a feel for whether they have a good understanding or not, and then I tend to talk to them about blood clots in general. This idea that we know that smaller blood clots that are, you know, potentially the edge of the lungs we don't really know what the right form of treatment is. And we use, you know historically, when you have a clot, but we don't know that's the right thing to do.

Clinician 1

For patients, the diagnosis of PE – especially when symptoms were not obvious – often brought significant anxiety, particularly around the possibility of not receiving anticoagulants. Many viewed participation in the trial as potentially withholding treatment, which heightened their concerns about taking part in the trial:

I guess it is worrying, obviously a doctor has just told you that you have blood clots on your lungs, regardless as to what size they might be. Obviously, I hadn't been selected at that point and at that point, you think, 'If I don't have medication, could I get worse?' Can't they not give me treatment? Being a human, you automatically think, 'Oh my gosh! I could die. I might not be able to breathe. It might affect something else. The blood clot might move.'

Patient 1, female

I had mixed emotions. I'm not going to lie ... anxietywise and as a human, you think, "Have I done the right thing? Should I have drugs or not?" Obviously, it's a worrying thing to have clots in your lungs.

Patient 6, female

Patients appeared to have opinions about the study arms they were allocated to, often because of their previous health experiences and other ongoing health issues. For example, patients' views on taking medication when they already had other health conditions and/or took medication influenced whether they thought being in the no anticoagulant therapy study arm was acceptable or not:

I wanted to be without the drugs completely (...) I'd had two years on quite high dose infusions for my disease. I'd been on really heavy drugs, and I just wanted to drugfree. For me, you know, I just didn't want to be given anymore drugs. I just didn't want to take anything else.

Patient 4, female

I think because I'm not someone who's ever very fortunately had to take any medication, I was particularly sort of hoping that I was going to end up in the side of the trial that was not [taking medication]. I think I was rather upset when I got allocated to the medication group. I wanted to get well quickly; you know.

Patient 8, male

I was selected to not have them [anticoagulants] this time and I was so thankful that I didn't have to. I remember weeks previously having the injections when I came out of surgery and just feeling so sore to thinking my body could potentially get rid of it and be healthier without the injections than with them, if I'm honest.

Patient 3, female

Despite these challenges, the involvement of clinicians in the consent and management process had a positive impact on how patients perceived and experienced study participation. Many patients reported feeling reassured by knowing they could contact clinical staff if they had concerns:

I did feel like I could go back to the hospital at any point and say, 'I don't feel well', or 'I don't feel that this is right for me. It's too much for me to handle', and I could have gone off the program, so I felt safe.

Patient 7, male

However, there were instances where conflicting medical advice added complexity to patients' experiences. One patient, who was initially assigned to the non-medication group, had their allocation overturned by another consultant due to concerns related to their existing condition:

I was randomised not to have medication. But then my other doctor consultant who deals with my [disease] overturned it because she felt that I needed to have the medication because of the blood clots, but also with the medication I'm on, there's a possibility I could produce more clots so she was more concerned about that. I was only concerned because of what the other consultant had told me originally about my [disease], I was only concerned about that. I was willing to do it without the medication but because of the concerns of my other doctor, they decided to put me on the medication anyhow.

Trust in clinicians and their decision-making was recognised as a crucial factor that likely influenced patients' decisions to participate in and continue with the trial, especially for those who were ultimately assigned to the no-anticoagulant group.

Discussion

The interview data with clinicians and patients, albeit limited by a small participant sample, emphasise the complex and overlapping challenges involved in changing established clinical practice, particularly in the context of managing PE and conducting research that impacts accepted clinical norms. Our findings reveal that issues such as clinician equipoise, clinician discomfort with certain patient characteristics, the need to reframe information provision and patient communication are intricately connected and relevant to both the challenge of changing clinical practice and the conduct of the trial. In this section, we integrate insights from related studies to offer a broader context for understanding these challenges. We emphasise the importance of nuanced approaches to implementing complex interventions that challenge established practices and recruiting for trials with varying intervention arms.

One of the key issues identified in our analysis is the role of clinician equipoise - or the lack thereof - in shaping decisions about patient care and recruitment. Our findings suggest that it was often challenging for clinicians to 'unwind' the information about PE that patients had received from other clinical staff and therefore convince them that anticoagulation may not be required. They often refrained from approaching eligible patients for study participation, especially when patients had already received a diagnosis and treatment plan from another clinician who believed anticoagulation was necessary. They also felt more uncomfortable recruiting particular groups of patients, such as those who were symptomatic. The findings also reflected broader difficulties in clinical practice, such as the challenge of communicating the potential risks and benefits of non-medication strategies for incidental subsegmental PE. Clinicians expressed discomfort when discussing these options with patients, largely due to concerns about potentially undermining the patient's trust in their care. Explaining the concept of equipoise to patients can be challenging, 11 as our findings indicated. Yet, patients can decide not to join a trial based on how they perceive their clinicians' treatment preferences.19

A number of other studies have reported similar findings, reinforcing the challenges we observed in our research. For instance, a lack of equipoise among clinicians has previously been recognised as a significant factor contributing to the low recruitment for trials. 9.19-21 Studies found that clinicians face many practical and emotional challenges as they work to reconcile the conflicts that arise between their clinical responsibilities and their research roles. 10,11 The concept of 'compulsion for diagnosis'22 reflects the deep-rooted tendency among clinicians to adhere to established diagnostic and treatment protocols. This need appears to be driven by a desire to provide patients with a sense of certainty and security, but it can also hinder the adoption of new practices that challenge the status quo.

Research on clinical decision-making has further highlighted the difficulty in balancing the need for clear evidence-based communication with the complexities of individual patient cases.^{23,24} The decision to administer or withhold anticoagulation involves substantial changes in care delivery, which can be challenging for clinicians to navigate, especially in the context of inherent uncertainty that accompanies such decisions.²⁵ Psychological mechanisms tied to the role of diagnosis further complicate these situations as patients often have strong expectations for clinicians to provide a diagnosis and treatment for their condition. These expectations can

influence clinical decision-making and present obstacles to recruiting patients for research trials.

Our findings underscore the complex interplay between patient perceptions, clinical decision-making and the management of PE, building on previous research that has highlighted patients' concerns about unknown side effects and uncertain treatment effectiveness in trial participation.^{26,27} In this study, patients diagnosed with PE, particularly those without overt symptoms, often experienced significant anxiety about the possibility of not receiving anticoagulants. This anxiety appeared to stem from a fear that the absence of medication could worsen their condition, reflecting the common concern that withholding treatment equates to increased risk. The role of clinicians in managing these anxieties proved crucial. In contrast to the clinicians who feared that such conversations might undermine patient trust or cause confusion, patients saw discussion of treatment as valuable as it allowed to express their concerns and reach an informed decision.²⁸

Research on recruitment to randomised trials has shown that when recruiters are actively involved in the consent process and maintain clear communication, patients tend to feel more secure and confident in their participation.²⁹⁻³¹ This reassurance is particularly important for those who are initially apprehensive about the trial,³² as it helps alleviate fears about being placed in less conventional treatment groups, such as the no-anticoagulant arm.

In conclusion, our analysis highlights the multifaceted challenges when attempting to conduct research that challenges accepted norms. The complexities of clinician equipoise, patient communication and the dynamics involved in diagnosis are deeply intertwined, influencing both clinical decision-making and patient recruitment for trials. These findings emphasise the need for nuanced approaches that address the overlapping concerns of clinicians and patients.

Implications for future research

Many of the challenges highlighted in the study are not unique but rather indicative of broader issues faced in delivering complex interventions that challenge established clinical practices and studies recruiting for trials with varying intervention arms. Below, we explore the key lessons learnt.

First, the difficulties related to clinicians' equipoise and obtaining consent for the study warrant further exploration. Our findings suggest that clinicians' decisions to approach and recruit patients were influenced by whether the PE diagnosis had been communicated by another member of the clinical team. This highlights the need to anticipate and address sources of uncertainty that may arise during the recruitment process. Providing additional support and training for clinicians to maintain equipoise and effectively communicate the risks and benefits of new clinical practices is essential for successful patient recruitment and trial participation. 10,11 Future research should focus on strategies to support clinicians in navigating these challenges.11

Second, recruitment issues are likely to remain a significant consideration in this type of research. Despite concerns that the no-anticoagulation arm might hinder patient recruitment, most patients found this aspect acceptable, even though they had distinct preferences for their assigned trial arm. Understanding patients' experiences of being approached and recruited is critical for improving strategies.²⁸ recruitment Future research explore the factors that influence patients' perceptions specifically for such interventions and trials to better tailor communication approaches and support recruitment efforts. Current recruitment and research processes may need to be adapted to meet the specific needs of trials that challenge accepted practices and where there is a lack of clinical equipoise. 11,33

Third, the practicalities of implementing the intervention must be carefully considered. Factors such as the timing of the trial and the challenges of restarting research after the COVID-19 pandemic likely impacted our recruitment rates. The complexity of the intervention itself may have influenced both patient and clinician experiences, contributing to the challenges reported. Future studies should carefully evaluate and address these practical challenges to ensure successful recruitment and implementation.

Finally, although the trial ceased prematurely, our findings suggest the need for further research into the long-term effects of the no-anticoagulant therapy. If patients, aware of their PE diagnosis, are not treated, this may alter their response to transient symptoms (e.g. leg or chest pain), potentially leading to increased healthcare utilisation, such as excess scans and emergency presentations. Future research should explore the psychological impact of not medicating and assess the potential 'harm' associated with repeated diagnostic imaging in this context.

Limitations

This study has several important limitations that should be noted. Our findings are based on the experiences of a limited number of patients and clinicians across a small number of study sites. As a result, the challenges reported may not be encountered or applicable outside the acute care setting. Additionally, our ability to gather more data and achieve thematic saturation was constrained by the premature termination of the trial. However, the challenges we identified are those that were most significant to the study participants and research team.

Furthermore, this study employed interviews with patients and clinicians as methods of data collection. Other methods, such as direct observations of clinical interactions and focus groups, could have provided additional perspectives and a more nuanced understanding of the challenges faced during trial recruitment and implementation.

Equality, diversity and inclusion

The STOP-APE randomised clinical trial was stopped prematurely due to very low recruitment, and we were unable to assess how well subjects recruited to this qualitative study represented the populations served at trial sites.

Conclusions

This study highlights the complex and overlapping challenges involved in changing established clinical practices, particularly in the context of managing PE and recruiting to trials that impact accepted clinical norms. The complexities surrounding clinician equipoise, patient communication and clinician discomfort with certain patient characteristics are intricately connected, significantly influencing both clinical decision-making and patient recruitment for trials. Future research should focus on developing strategies to help clinicians maintain equipoise and communicate the risks and benefits of these interventions effectively. Moreover, gaining a deeper understanding of patients' experiences and perceptions is vital for enhancing recruitment strategies and boosting trial participation.

Additional information

CRediT contribution statement

Agnieszka Ignatowicz (https://orcid.org/0000-0002-5863-0828): Methodology, Formal analysis, Investigation, Resources, Writing – review and editing.

Sheila Greenfield (https://orcid.org/0000-0002-8796-4114): Methodology, Formal analysis, Investigation, Resources, Writing – review and editing.

Pooja Gaddu (https://orcid.org/0009-0009-4980-1689): Project administration, Writing – review and editing.

Clare Prince: Writing - review and editing.

Mark Toshner (https://orcid.org/0000-0002-3969-6143): Conceptualisation, Writing – review and editing.

Graham Robinson (https://orcid.org/0000-0001-5191-1451): Conceptualisation, Writing – review and editing.

Jonathan Rodrigues (https://orcid.org/0000-0003-2744-3509): Conceptualisation, Writing – review and editing.

Susan Jowett (https://orcid.org/0000-0001-8936-3745): Conceptualisation, Writing – review and editing.

Simon Noble (https://orcid.org/0000-0001-5425-2383): Conceptualisation, Writing – review and editing.

Michael Newnham (https://orcid.org/0000-0002-2972-6249): Conceptualisation, Writing – review and editing.

Alice Turner (https://orcid.org/0000-0002-5947-3254): Conceptualisation, Writing – review and editing.

Daniel Lasserson (https://orcid.org/0000-0001-8274-5580): Conceptualisation, Methodology, Writing – review and editing, Funding acquisition.

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/PSDG7298.

Primary conflicts of interest: Agnieszka Ignatowicz has received grants from NIHR Genomic Medicine Service Alliance (GMSA), NIHR Global Health Research (GHR) and NIHR HTA Programmes.

Mark Toshner is funded by the NIHR Cambridge BRC and reports consulting fees from Jansen and MorphogenIX and support for attending meetings from Glaxo Smith Kline and Jansen. He participates on boards for FluCov and ComCov.

Jonathan Rodrigues has received funding from the NIHR Research for Patient Benefit Programme and reports consulting fees from NHS X and Heartflow, speaker fees from Sanofi and Aidence, is the Director of Clinical Quality at Heart and Lung Imaging Limited and declares shares in Radnet.

Susan Jowett was a member of the NIHR NTA CET 2016–20 and HTA Funding Committee 2019–21.

Simon Noble has received honoraria from Leo Pharma and Pfizer. He is Chair of MePFAC Trial Steering Committee, Database Safety and Monitoring committee member for Prepare Kidney and Medical Director of Thrombosis UK.

Alice Turner is a member of the NIHR HTA prioritisation committee (2020–5). Outside this work/subject area, she has received research funds and/or honoraria from Vertex, AstraZeneca, CSL Behring, Grifols Biotherapeutics, GSK, Chiesi, Phillips, ResMed and Boehringer within the last 3 years.

Daniel Lasserson is funded by the NIHR Applied Research Collaboration (ARC) West Midlands, the NIHR Community Healthcare MedTech and IVD Cooperative (MIC) and the NIHR Oxford Biomedical Research Centre (BRC), and he has received funding from the NIHR Policy Research Programme. He was on the NIHR HTA CET Committee 2016–22. He reports institutional funding from Vifor Pharma Ltd and Butterfly Net Inc. for research projects outside this study.

Data-sharing statement

There are no additional data to share for this analysis of the STOP-APE trial. Due to the very small number of patients recruited and closure of the trial, all relevant data are presented here, and there are no data that can be shared.

Ethics statement

Research ethics approval was given on 18 September 2020 by Wales Research Ethics Committee 6. REC Reference: 20/WA/026.

Information governance statement

The University of Birmingham is committed to handling all personal information in line with the UK Data Protection Act (2018) and the General Data Protection Regulation (EU GDPR) 2016/679. Under the Data Protection legislation, the University of Birmingham is the Data Controller, and you can find out more about how we handle personal data, including how to exercise your individual rights and the contact details for our Data Protection Officer here: The Data Protection Office, Legal Services, University of Birmingham, Edgbaston, Birmingham, B15 2TT. E-mail: dataprotection@contacts.bham.ac.uk Telephone: 0121 414 3916.

Department of Health and Social Care disclaimer

This publication presents independent research commissioned by the National Institute for Health and Care Research (NIHR).

The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, MRC, NIHR Coordinating Centre, the Health Technology Assessment programme or the Department of Health and Social Care.

This article was published based on current knowledge at the time and date of publication. NIHR is committed to being inclusive and will continually monitor best practice and guidance in relation to terminology and language to ensure that we remain relevant to our stakeholders.

Study registration

The trial was prospectively registered as ISRCTN15645679.

Funding

This article presents independent research funded by the National Institute for Health and Care Research (NIHR) Health Technology Assessment programme as award number NIHR128073.

This article reports on one component of the research award Stopping anticoagulation for isolated or incidental sub-segmental pulmonary embolism (STOPAPE). For more information about this research please view the award page https://fundingawards.nihr.ac.uk/award/NIHR128073

About this article

The contractual start date for this research was in December 2019. This article began editorial review in September 2023 and was accepted for publication in October 2024. The authors have been wholly responsible for all data collection, analysis and interpretation and for writing up their work. The Health Technology Assessment editors and publisher have tried to ensure the accuracy of the authors' article and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this article.

Copyright

Copyright © 2025 Ignatowicz *et al.* This work was produced by Ignatowicz *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open-Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution, the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

List of abbreviations

NIHR National Institute for Health and

Care Research

PE pulmonary embolism

References

- Myall M, May C, Richardson A, Bogle S, Campling N, Dace S, Lund S. Creating pre-conditions for change in clinical practice: the influence of interactions between multiple contexts and human agency. *J Health Organ Manag* 2020;35:1–17. https://doi.org/10.1108/jhom-06-2020-0240
- 2. Gupta DM, Boland RJ, Aron DC. The physician's experience of changing clinical practice: a struggle to unlearn. *Implement Sci* 2017;**12**:28. https://doi.org/10.1186/s13012-017-0555-2
- 3. van Bodegom-Vos L, Davidoff F, Marang-van de Mheen PJ. Implementation and de-implementation: two sides of the same coin? *BMJ Qual Saf* 2017;**26**:495–501. https://doi.org/10.1136/bmjqs-2016-005473
- 4. Cochrane LJ, Olson CA, Murray S, Dupuis M, Tooman T, Hayes S. Gaps between knowing and doing: understanding and assessing the barriers to optimal health care. *J Contin Educ Health Prof* 2007;**27**:94–102. https://doi.org/10.1002/chp.106
- Cabana MD, Rand CS, Powe NR, Wu AW, Wilson MH, Abboud PA, Rubin HR. Why don't physicians follow clinical practice guidelines? A framework for improvement. *JAMA* 1999;282:1458–65. https://doi. org/10.1001/jama.282.15.1458
- 6. Mather M, Pettigrew LM, Navaratnam S. Barriers and facilitators to clinical behaviour change by primary care practitioners: a theory-informed systematic review of reviews using the Theoretical Domains Framework and Behaviour Change Wheel. *Syst Rev* 2022;**11**:180. https://doi.org/10.1186/s13643-022-02030-2
- PateyAM, GrimshawJM, Francis JJ. Changing behaviour, 'more or less': do implementation and de-implementation interventions include different behaviour change techniques? *Implement Sci* 2021;16:20. https://doi. org/10.1186/s13012-021-01089-0
- 8. Colquhoun HL, Squires JE, Kolehmainen N, Fraser C, Grimshaw JM. Methods for designing interventions to change healthcare professionals' behaviour: a systematic review. *Implement Sci* 2017;**12**:30. https://doi.org/10.1186/s13012-017-0560-5
- 9. Fayter D, McDaid C, Eastwood A. A systematic review highlights threats to validity in studies of barriers to cancer trial participation. *J Clin Epidemiol*

- 2007;**60**:990–1001. https://doi.org/10.1016/j. jclinepi.2006.12.013
- Donovan JL, de Salis I, Toerien M, Paramasivan S, Hamdy FC, Blazeby JM. The intellectual challenges and emotional consequences of equipoise contributed to the fragility of recruitment in six randomized controlled trials. *J Clin Epidemiol* 2014;67:912–20. https://doi.org/10.1016/j.jclinepi.2014.03.010
- Rooshenas L, Elliott D, Wade J, Jepson M, Paramasivan S, Strong S, et al.; ACST-2 study group. Conveying equipoise during recruitment for clinical trials: qualitative synthesis of clinicians' practices across six randomised controlled trials. PLOS Med 2016;13:e1002147. https://doi.org/10.1371/journal.pmed.1002147
- 12. Lasserson D, Gaddu P, Mehta S, Ignatowicz A, Greenfield S, Prince C, et al. Stopping anticoagulation for isolated or incidental pulmonary embolism: the STOPAPE RCT protocol. *Health Technol Assess* 2024:1–17. https://doi.org/10.3310/HRCW7937
- Duffett L, Castellucci LA, Forgie MA. Pulmonary embolism: update on management and controversies. BMJ 2020;370:m2177. https://doi.org/10.1136/bmj. m2177
- 14. Auer RC, Schulman AR, Tuorto S, Gönen M, Gonsalves J, Schwartz L, et al. Use of helical CT is associated with an increased incidence of postoperative pulmonary emboli in cancer patients with no change in the number of fatal pulmonary emboli. J Am Coll Surg 2009;208:871–8; discussion 878. https://doi.org/10.1016/j.jamcollsurg.2008.12.030
- 15. Goldhaber SZ, Grasso-Correnti N. Treatment of blood clots. *Circulation* 2002;**106**:e138–40. https://doi.org/10.1161/01.CIR.0000038923.61628.3D
- McGill E, Marks D, Er V, Penney T, Petticrew M, Egan M. Qualitative process evaluation from a complex systems perspective: a systematic review and framework for public health evaluators. *PLOS Med* 2020;17:e1003368. https://doi.org/10.1371/journal. pmed.1003368
- Khan TH, MacEachen E. An alternative method of interviewing: critical reflections on videoconference interviews for qualitative data collection. *Int J Qual Methods* 2022;**21**:16094069221090063. https://doi. org/10.1177/16094069221090063
- 18. Gale NK, Heath G, Cameron E, Rashid S, Redwood S. Using the framework method for the analysis of qualitative data in multi-disciplinary health research. *BMC Med Res Methodol* 2013;13:117. https://doi.org/10.1186/1471-2288-13-117
- 19. Mills N, Donovan JL, Smith M, Jacoby A, Neal DE, Hamdy FC. Perceptions of equipoise are crucial to trial participation: a qualitative study of men in the ProtecT

- study. Control Clin Trials 2003;24:272-82. https://doi. org/10.1016/s0197-2456(03)00020-5
- 20. Cook JA, McCulloch P, Blazeby JM, Beard DJ, Marinac-Dabic D, Sedrakyan A; IDEAL Group. IDEAL framework for surgical innovation 3: randomised controlled trials in the assessment stage and evaluations in the long term study stage. BMJ 2013;346:f2820. https://doi.org/10.1136/bmj.f2820
- 21. Miller FG. Joffe S. Equipoise and the dilemma of randomized clinical trials. N Engl J Med 2011;364:476-80. https://doi.org/10.1056/NEJMsb1011301
- 22. Coon ER, Quinonez RA, Moyer VA, Schroeder AR. Overdiagnosis: how our compulsion for diagnosis may be harming children. Pediatrics 2014;134:1013-23. https://doi.org/10.1542/peds.2014-1778
- 23. Engebretsen E, Heggen K, Wieringa S, Greenhalgh T. Uncertainty and objectivity in clinical decision making: a clinical case in emergency medicine. Med Health Care Philos 2016;19:595-603. https://doi.org/10.1007/ s11019-016-9714-5
- 24. Greenhalgh T, Kostopoulou O, Harries C. Making decisions about benefits and harms of medicines. 2004;329:47-50. https://doi.org/10.1136/ bmj.329.7456.47
- 25. DuBose BM, Mayo AM. Resistance to change: a concept analysis. Nurs Forum 2020;55:631-6. https://doi. org/10.1111/nuf.12479
- 26. Sheridan R, Martin-Kerry J, Hudson J, Parker A, Bower P. Knapp P. Why do patients take part in research? An overview of systematic reviews of psychosocial barriers and facilitators. Trials 2020;21:259. https:// doi.org/10.1186/s13063-020-4197-3
- 27. Prescott RJ, Counsell CE, Gillespie WJ, Grant AM, Russell IT, Kiauka S, et al. Factors that limit the quality,

- number and progress of randomised controlled trials. Health Technol Assess 1999;3:1-143.
- 28. Mills N, Donovan JL, Wade J, Hamdy FC, Neal DE, Lane JA. Exploring treatment preferences facilitated recruitment to randomized controlled trials. Clin Epidemiol 2011;64:1127-36. https://doi. org/10.1016/j.jclinepi.2010.12.017
- 29. McDonald CE, Paynter C, Francis JJ, Rodda D, Bajwa S, Jackson D, Story D. Exploring patient acceptability of a short-stay care pathway in hospital post arthroplasty: a theory-informed qualitative study. Health Expect 2022;25:2002-14. https://doi.org/10.1111/ hex.13561
- 30. Ahalt C, Walter LC, Yourman L, Eng C, Pérez-Stable EJ, Smith AK. 'Knowing is better': preferences of diverse older adults for discussing prognosis. J Gen Intern 2012;**27**:568-75. https://doi.org/10.1007/ s11606-011-1933-0
- 31. Ross S, Grant A, Counsell C, Gillespie W, Russell I, Prescott R. Barriers to participation in randomised controlled trials: a systematic review. J Clin Epidemiol 1999;52:1143-56. https://doi.org/10.1016/ 50895-4356(99)00141-9
- 32. Stryker JE, Wray RJ, Emmons KM, Winer E, Demetri G. Understanding the decisions of cancer clinical trial participants to enter research studies: factors associated with informed consent, patient satisfaction, and decisional regret. Patient Educ Couns 2006;63:104-9. https://doi.org/10.1016/j.pec.2005.09.006
- 33. Farrar N, Elliott D, Houghton C, Jepson M, Mills N, Paramasivan S, et al. Understanding the perspectives of recruiters is key to improving randomised controlled trial enrolment: a qualitative evidence synthesis. Trials 2022;23:883. https://doi.org/10.1186/ s13063-022-06818-4